Author’s response to reviews

Title: Feasibility of provision of sustainable intra-dermal post exposure prophylaxis against rabies at primary care level - evidence from rural Haryana, India

Authors:

Harshal Salve Dr (harshalsalve@ymail.com)
Sanjeev Kumar Dr (docsanjiv@gmail.com)
Rizwan SA Dr (sarizwan1986@gmail.com)
Sanjay K Rai Dr (drsanjay.aiims@gmail.com)
Shashi Kant Dr (skant76@gmail.com)
Chandrakant S Pandav Dr (cpandav@iqplusin.org)

Version: 2 Date: 14 October 2013

Author’s response to reviews:

I have uploaded cover letter